Author's response to reviews

Title: DIABRISK - SL Prevention of cardio-metabolic disease with life style modification in young urban Sri Lankan's - study protocol for a randomized controlled trial

Authors:

Mahen Wijesuriya (amrit@slt.lk)
Martin Gulliford (martin.gulliford@kcl.ac.uk)
Laksha Vasantharajah (dasl@sltnet.lk)
Giancarlo Viberti (andrea.collins@kcl.ac.uk)
Luigi Gnudi (luigi.gnudi@kcl.ac.uk)
Janaka Karalliedde (j.karalliedde@kcl.ac.uk)

Version: 11 Date: 22 September 2011

Author's response to reviews:

Figure title and legend have been added as requested after the references in revised manuscript.